

The risks of using non-specific outcome measures to capture activities of daily living in myotonic dystrophy type 2

Citation for published version (APA):

Hamadeh, T., Bovenkerk, D. S. H., Faber, C. G., & Merkies, I. S. J. (2021). The risks of using non-specific outcome measures to capture activities of daily living in myotonic dystrophy type 2. *Neuromuscular Disorders*, *31*(4), 367-368. https://doi.org/10.1016/j.nmd.2021.02.008

Document status and date: Published: 01/04/2021

DOI: 10.1016/j.nmd.2021.02.008

Document Version: Publisher's PDF, also known as Version of record

Document license: Taverne

Please check the document version of this publication:

• A submitted manuscript is the version of the article upon submission and before peer-review. There can be important differences between the submitted version and the official published version of record. People interested in the research are advised to contact the author for the final version of the publication, or visit the DOI to the publisher's website.

• The final author version and the galley proof are versions of the publication after peer review.

• The final published version features the final layout of the paper including the volume, issue and page numbers.

Link to publication

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
- You may not further distribute the material or use it for any profit-making activity or commercial gain
- You may freely distribute the URL identifying the publication in the public portal.

If the publication is distributed under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license above, please follow below link for the End User Agreement:

www.umlib.nl/taverne-license

Take down policy

If you believe that this document breaches copyright please contact us at:

repository@maastrichtuniversity.nl

providing details and we will investigate your claim.



Available online at www.sciencedirect.com

ScienceDirect

Neuromuscular Disorders 31 (2021) 367-368



Letter to the Editor

The risks of using non-specific outcome measures to capture activities of daily living in myotonic dystrophy type 2

We read with great interest the recently published study by colleagues Montagnese et al. titled "How to capture activities of daily living in myotonic dystrophy type 2?". In this study, the authors analyzed the performance of outcome measures that are specifically designed for other illnesses in a group of myotonic dystrophy type 2 (DM 2) patients [1]. It has been described in the literature how factors such as item weighting and relevance in outcome measures vary significantly between different illnesses. For this reason, we consider that the risks embedded in their conclusions are significant and worth acknowledging.

In this modern scientific era, valid and reliable outcome measures are necessary to accurately monitor the limitations in daily life activities and social participation, the progress of a disease, and its response to treatment. Ordinal scales, which provide a systemic ordering without a true numerical value, are widely used in clinical practice. However, these scales are prone to differential sensitivity, making them inapplicable for parametric statistical testing and inadequate for an accurate interpretation of clinical trial results [2,3]. Additionally, classical test theory ordinal-based metrics may include items that are not relevant to the patient's ability and may achieve a total sum score that incorrectly assumes equal weight and relevance of each item [2,4]. For healthrelated evaluation measurements, there are modern scientific techniques at our disposal. The Rasch analysis, for example, evaluates the probability of item completion depending on the item difficulty and the patient's ability. This process transforms these scales into linearly weighted "interval" measures and improves the quality of outcome measures and interpretation of the results of clinical trials [4,5].

Based on this concept, the R-PAct and DM1-Activ-c scales were constructed [6,7]. The R-PAct (Rasch-built Pompe-specific activity scale) was validated in a study population of 186 patients with Pompe disease in 2013 [6]. The Rasch-built DM1-Activ-c (Myotonic dystrophy type 1 activity and participation scale) was reconstructed in 2015 using 312 records of genetically confirmed DM 1 patients [7]. A fundamental fact is that myotonic dystrophies type 1 and 2 (DM 1, DM 2) are two clinically, histopathologically and genetically distinct disorders while sharing the same

eponym considering their common symptoms of myotonia, muscular weakness, and muscular atrophy. DM 1 has a predominant distal weakness distribution; a congenital, juvenile, and adult form have been described, it has a higher frequency of respiratory failure, more facial weakness, a high predominance of myotonia, complaints of myalgia are rare, and it has a worse general prognosis. In contrast, DM 2 has a lower world-wide frequency, there is a predominantly proximal and axial muscular involvement, onset occurs during adulthood, respiratory involvement is rare, myotonia is less prominent, myalgia has a higher prevalence, and generally, a better prognosis is documented [8,1].

The article under discussion, aimed to evaluate the validity and performance of the aforementioned scales, R-PAct and DM1-Activ-c, among others, in a cohort of genetically confirmed DM 2 patients [1]. R-PAct was included because of the similarity between Pompe and DM 2 regarding muscle weakness distribution. They tested and compared their performance and found during the evaluation of DM1-Activc a higher ceiling effect in DM 2 patients in comparison to the DM 1 cohort, a maintained trend of increasing difficulty of the scale, and a moderate difference in the hierarchy of item-difficulty. No DM 2 patient scored low enough to indicate severe limitations and there was no difference observed between men and women. Additionally, there were no significant differences observed in scores over time, corroborating a slower disease progression of DM 2 and at the same time, the need for studies with a longer follow-up for a better assessment of the responsiveness in DM 2. [1]

R-PAct also had a higher ceiling effect (14.7%) that significantly differed from the results found in Late-Onset Pompe Disease patients (0.4%), the trend of increasing difficulty was maintained, and only minor changes were detected in the hierarchy of item difficulty. The low impact of myotonia in both scales was confirmed by its weak correlation coefficient, and with the use of regression analysis, they found that the major predictors of lower scores were higher age, longer disease duration, shorter 6MWT distance, and presence of myalgia. The authors concluded that DM1-Activ-c might perform slightly better than R-PAct in depicting the burden of DM 2 patients in activities of daily living (ADLs) and could be adopted for monitoring disease progression in DM 2 patients until a disease-specific patient-reported outcome measure (PROM) is available [1].

T. Hamadeh, D.S.H. Bovenkerk, C.G. Faber et al.

It must be noted that this conclusion poses a risk of misinterpretation of data since it's not certain that the scales' validity can be extrapolated to another disease with different characteristics and clinical manifestations. Additionally, the use of a disease-specific scale does not guarantee that it will adequately capture the entire severity spectrum of another disorder. The increase of the ceiling effect found when evaluating the performance of DM1-Activ-c (11.6% in DM 2 vs. 7.6% in DM 1) which was attributed by the authors to a milder impairment of ADLs in DM 2, could also be explained by a possible ineffectiveness of the items to capture the most severe presentations of DM 2.

In the Rasch-built overall disability scale for immunemediated peripheral neuropathies described in 2011 by van Nes et al., Multifocal motor neuropathy (MMN) patients were initially included assuming a pathophysiological overlap but a significantly different behaviour in functional response to tasks was observed in comparison with the cohort of patients diagnosed with Guillain-Barré Syndrome (GBS), Chronic Inflammatory Demyelinating Polyneuropathy (CIDP) and immunoglobulin M monoclonal gammopathy of undetermined significance associated polyneuropathy (MGUSP) [4]. For MMN patients, fine motor upper limb items were considered much more difficult to accomplish than mobility items (e.g., walking, standing, and running). This supports the argument that for different illnesses, item selection and item difficulty are not always comparable. This observation led to the eventual exclusion of MMN patients from the study, but also to the later construction of an MMN-specific scale in 2015 [9].

Therefore, it is of uttermost importance to develop validated disease-specific measurement instruments to arrive to correct conclusions and avoid misinterpretations. It's uncertain that we can safely use one scale validated for one disease to measure limitations in patients with another disorder, as items that are relevant for patients with a proximal myopathy do not necessarily apply to patients suffering from another disease with similar muscle weakness distribution, or to patients with a predominant distal muscle involvement [10]. A substantial cohort of patients could be assembled using available multicenter DM 2 databases in addition to the existing DM 2 data gathered by Montagnese et al. to develop a validated DM2-specific scale that can overcome these limitations. This new scale could initially include the same items that are found in DM1-Activ-c, but a different weighting and relevance of the items are to be expected.

References

- Montagnese F, Rastelli E, Stahl K, Massa R, Schoser B. How to capture activities of daily living in myotonic dystrophy type 2? Neuromuscul Disord 2020;30:796–806.
- [2] DeVellis RF. Classical test theory. Med Care 2006;44:S50-9.
- [3] Wright BD, Linacre JM. Observations are always ordinal; measurements, however, must be interval. Arch Phys Med Rehabil 1989;70:857–60.
- [4] van Nes SI, Vanhoutte EK, van Doorn PA, et al. Rasch-built overall disability scale (R-ODS) for immune-mediated peripheral neuropathies. Neurology 2011;76:337–45.
- [5] Tennant A, Conaghan PG. The Rasch measurement model in rheumatology: what is it and why use it? When should it be applied, and what should one look for in a Rasch paper? Arthritis Rheum 2007;57:1358–62.
- [6] van der Beek NA, Hagemans ML, van der Ploeg AT, van Doorn PA, Merkies IS. The Rasch-built Pompe-specific activity (R-PAct) scale. Neuromuscul Disord 2013;23:256–64.
- [7] Hermans MC, Hoeijmakers JG, Faber CG, Merkies IS. Reconstructing the Rasch-built myotonic dystrophy type 1 activity and participation scale. PLoS ONE 2015;10:e0139944.
- [8] Wenninger S, Montagnese F, Schoser B. Core clinical phenotypes in myotonic dystrophies. Front Neurol 2018;9:303.
- [9] Vanhoutte EK, Faber CG, van Nes SI, et al. Rasch-built overall disability scale for multifocal motor neuropathy (MMN-RODS(©)). J Peripher Nerv Syst 2015;20:296–305.
- [10] Mul K, Horlings CGC, Faber CG, van Engelen BGM, Merkies ISJ. Rasch analysis to evaluate the motor function measure for patients with facioscapulohumeral muscular dystrophy. Int J Rehabil Res 2020;44(1):38–44.

Tatiana Hamadeh* David S.H. Bovenkerk Catharina G. Faber Department of Neurology, Maastricht University Medical Center, P.O. Box 5800, Maastricht, the Netherlands

Ingemar S.J. Merkies Department of Neurology, Maastricht University Medical Center, P.O. Box 5800, Maastricht, the Netherlands Department of Neurology, Curacao Medical Center, J.H.J. Hamelbergweg z/n, Willemstad, Curaçao

*Corresponding author.

E-mail address: t.hamadeh@maastrichtuniversity.nl (T. Hamadeh)